
diff -r 64c20ed7941a -r b1c2f5b85880 handouts/ho02.tex
--- a/handouts/ho02.tex	Thu Feb 11 09:33:01 2016 +0000
+++ b/handouts/ho02.tex	Thu Feb 11 09:41:31 2016 +0000
@@ -119,7 +119,7 @@
 \end{itemize}
 
 \noindent If you are interested in the recent state of affairs
-of e-voting machinery, I recommend the talk Jeremy Epstein
+of e-voting machinery, I recommend a talk by Jeremy Epstein
 
 \begin{center}
 \url{https://www.usenix.org/sites/default/files/conference/protected-files/jets15_slides_epstein.pdf}




